University of Toronto, Department of Electrical and Computer Engineering

ECE241F - Information Rele vant to Mid-Term

Fall 1999 A. Carusone, P. Lehn, J. Rose Midterm Date and Time: Thursda y October 14, from 6-8pm Midterm Location - Depending on Y our Last Name

The Haultain building is located east of Mechanical Engineering.

Ramsey Wright is at the corner of south-west corner of St. George and Harbord. Those two rooms are at the east end of the building in the the structure right at the south-west corner of St. George and Harbord.

New College, Wilson Hall is at the corner of Spadina and Willcocks. See http://oracle.osm.utoronto.ca/map/ for a map.

Sections of the T ext That Have Been Co vered & You’re Responsib le for Chapter 1, Chapter 2 (except section 2.5.1), Chapter 3, Section 3.1-3.5, 3.6.2, 3.6.3, 3.6.4, pages 123-129 of 3.10, Chapter 4: 4.1, 4.2, 4.3, 4.5, 4.6 and 4.7.1 And, of course, you’re responsible for everything covered in labs 1-3.

Assigned Pr oblems to Do From Text

Chapter 2: 2.2, 2.19, 2.21, 2.24, 2.25 Chapter 3: 3.1, 3.6, 3.8 and 3.12. Chapter 4: 4.2, 4.6, 4.11, 4.12, 4.18 Solutions to Assigned Pr oblems

Were handed out in hard copy only, in class. They are not available online.

Last Name Range Location A to Chow Haultain Building Room HA 410 Chu to Kingston Ramsey Wright Building Room RW 110 Kok to M Ramsey Wright Building Room RW 117 N to Thiffault New College, Wilson Hall, Room WI 1016 Thompson to Z New College, Wilson Hall, Room WI 1017